Author's response to reviews

Title: Acute oxygenation changes on ischemic foot of a novel intermittent pneumatic compression device and of an existing sequential device in severe peripheral arterial disease.

Authors:

   Fabio Manfredini (mdf@unife.it)
   Anna Maria Malagoni (mlgnmr@unife.it)
   Michele Felisatti (flsmhl@unife.it)
   Simona Mandini (mndsmn@unife.it)
   Nicola Lamberti (lmbncl@unife.it)
   Roberto Manfredini (mfr@unife.it)
   Francesco Mascoli (msf@unife.it)
   Nino Basaglia (n.basaglia@ospfe.it)
   Paolo Zamboni (zambo@unife.it)

Version: 4 Date: 24 March 2014

Author's response to reviews: see over
Dear Editor,

Thank you for considering our manuscript suitable for publication in BMC Cardiovascular Disorders.

The editorial revisions requested have been made, as follows:

i) The full name of the institutional review board that approved the study has been stated (P5, L6).

ii) “Perspectives” has been removed from the heading (P13, L19).

iii) Authors’ Contribution section has been moved further up so that it follows the Conflict of Interest section (P14).

The new version of the manuscript has been submitted with changes highlighted by MS Word tracked changes.

Kind regards

Anna Maria Malagoni

Address for correspondence:

Anna Maria Malagoni, MD, PhD

Program Pathophysiology of Vascular Peripheral System and Day Surgery, S. Anna Hospital

University of Ferrara

Via Aldo Moro 8

44124 Ferrara-Loc. Cona, Italy

Tel +39 0532 239783

Fax +39 0532 237443

E-mail: mlgnmr@unife.it